Author’s response to reviews

Title: BRAF mutant colorectal cancer: ErbB2 expression levels as predictive factor for the response to combined BRAFi/ErbB1 inhibitors

Authors:

Evelina Miele (evelina.miele@opbg.net)
Luana Abballe (luanaabballe@uniroma1.it)
Gian Paolo Spinelli (gianpaolo.spinelli@uniroma1.it)
Zein Mersini Besharat (zeinmersinibesharat@uniroma1.it)
Giuseppina Catanzaro (giuseppinacatanzaro@uniroma1.it)
Martina Chiacchiarini (martinachiacchiarini@uniroma1.it)
Alessandra Vacca (alessandravacca@uniroma1.it)
Agnese Po (agnesepo@uniroma1.it)
Carlo Capalbo (carlocapalbo@uniroma1.it)
Elisabetta Ferretti (elisabettaferretti@uniroma1.it)

Version: 3 Date: 24 Jan 2020

Author’s response to reviews:

Point by Point response to Editor Comments:
1. Availability of data and material

The Availability of data and materials section refers to the raw data used in your study. By presenting tables and figures, it is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

Authors' reply: We indicated in the appropriate section that

• The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.
2. Clean copy
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors' reply: as requested we uploaded a clean version of the manuscript.

Finally, we have stylistically rephrased the last sentence in the abstract and corrected few typos in the text.

Thank you for your assistance

Best regards,
Evelina Miele